Author's response to reviews

Title: Cognitive performance in relapsing remitting multiple sclerosis: a longitudinal study in daily practice using a brief computerized cognitive battery

Authors:

Chris Edgar (chris.edgar@unitedbiosource.com)
Peter J Jongen (ms4ri@kpnmail.nl)
Evert Sanders (ResearchNeurologie@amphia.nl)
Christian Sindic (Christian.Sindic@uclouvain.be)
Sophie Goffette (docteurgoffette@yahoo.com)
Michel Dupuis (mi.dupuis@clinique-saint-pierre.be)
Philippe Jacquerye (p.jacquerye@clinique-saint-pierre.be)
Daniel Guillaume (daniel.guillaume@cnrf.be)
Regine Reznik (regine.reznik@cnrf.be)
Keith Wesnes (Keith.Wesnes@unitedbiosource.com)

Version: 14 Date: 6 June 2011

Author's response to reviews: see over
Revised format 060611

MS ID : 6189727054917604

Author list: Chris Edgar, Peter J Jongen, Evert Sanders, Christian Sindic, Sophie Goffette, Michel Dupuis, Philippe Jacquerye, Daniel Guillaume, Regine Reznik and Keith Wesnes

Title : Cognitive performance in relapsing remitting multiple sclerosis: a longitudinal study in daily practice using a brief computerized cognitive battery

Journal: BMC Neurology

Major revisions

Tables: table 7 has not been cited within the main text 'insert table here' notes do not count as citations. Please cite table 7.

Our apologies for not having updated the table numbering in the text correctly. The table numbering has been changed. Please see tracked changes in the manuscript on page 11, page 12, page 13, page 15 and page 17. Now table 7 is cited in the text (page 13).

Minor revisions

Tables: please remove 'insert table x here' notes.

The all 'insert table x' notes have been removed.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimize white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

We can’t crop the figures any further, so we have cut the figure and legend so they are separate pictures. Hopefully this is useful to you.

Additional files: please remove all additional files APART from files 1 and 2.

All additional files apart from files 1 and 2 have been removed.

Kind regards,

On behalf of the authors,

Peter J Jongen, neurologist